\documentclass[11pt]{article}
\usepackage{amsmath,amssymb,amsthm,enumitem,clrscode3e}

\title{\bf Midterm Corrections \\[2ex]
\rm\normalsize Due: February 25, 2020}
\date{}
\author{}

\begin{document}
\maketitle

\begin{center}
    \fbox{\fbox{\parbox{\textwidth}{\centering
    Your solutions must be typed (preferably typeset in \LaTeX ) and submitted as a hard-copy at the beginning of
    class on the day its due.
    }}}
\end{center}

\paragraph{Guidelines}
Choose up to three exam questions where you lost at least half the maximum possible points, each part of multipart question would count as a separate question. You may submit corrected answers for each of the chosen questions. These corrected answers will be worth 60\% of the maximum possible score for that question, with {\em no partial credit} being given. This score on each question will replace the original value from the midterm. Your answers are expected to contain more detail than what would have been accepted on exam day. As a general guideline, you should try to craft your answers as if you were creating an answer key for the test.

\paragraph{}
Unlike homework assignments, you must work on this independently. Feel free to do any external research you need to, but the final submission must be entirely a product of your own work.

\end{document}
